Author's response to reviews

Title: Identifying the needs of penile cancer sufferers: A systematic review of the quality of life, psychosexual and psychosocial literature in penile cancer.

Authors:

Satish B Maddineni (satishandsian@yahoo.co.uk)  
Maurice W Lau (Maurice.Lau@christie.nhs.uk)  
Vijay K Sangar (vijay.sangar@nhs.net)

Version: 3 Date: 5 August 2009

Author's response to reviews:

The references (5,6,7) have been put in the correct chronilogical order.  
Reference 15 has been fully cited with a weblink.  
A tables section has been added at the end of the manuscript as requested.  
The word title and the keywords have been removed as requested.  
Qualifications have been removed.  
Underlining has been removed.